Author’s response to reviews

Title: Design of the DIRECT-project: interventions to increase job resources and recovery opportunities to improve job-related health, well-being, and performance outcomes in nursing homes

Authors:

Ellen MB Spoor (e.m.b.spoor@tue.nl)
Jan De Jonge (j.d.jonge@tue.nl)
Jan PH Hamers (j.p.h.jamers@zw.unimaas.nl)

Version: 3 Date: 18 May 2010

Author’s response to reviews:

May 18th, 2010

Ms. Norton, MD
Editor-in-Chief
BMC Public Health

Dear Ms. Norton

Many thanks for the editorial descisions. Enclosed you will find the resubmission of our manuscript entitled “Design of the DIRECT-project: interventions to increase job resources and recovery opportunities to improve job-related health, well-being, and performance in nursing homes” (MS ID 1532920598386321).

Below you find the changes that we made:

• I revised the authors’ contribution section.
• I revised figure 2 and figure 4 because I discovered some small mistakes.
• I checked my manuscript for any typographical errors.

Thank you for considering our revised manuscript.

Yours sincerely, also on behalf of the co-authors (*),

Ellen Spoor

Eindhoven University of Technology
Department of Industrial Engineering and Innovation Sciences
Human Performance Management Group
P.O. Box 513, 5600 MB Eindhoven
The Netherlands
Email: e.m.b.spoor@tue.nl
Phone: 0031 40 247 5211
Fax: 0031 40 243 7161

(*) co-authors:
Jan de Jonge, Eindhoven University of Technology, Eindhoven, the Netherlands
Jan Hamers, Maastricht University, Maastricht, the Netherlands